Author's response to reviews

Title: The 'placement' of people with profound Impairments across the lifespan: re-thinking age criteria

Authors:

Barbara E Gibson (barbara.gibson@utoronto.ca)
Gillian King (gking27@uwo.ca)
Shauna Kingsnorth (skingsnorth@hollandbloorview.ca)
Patricia McKeever (p.mckeever@utoronto.ca)

Version: 4 Date: 10 April 2014

Author's response to reviews: see over
RE: Response to Reviews
The 'placement' of people with profound Impairments across the lifespan: still life
Gibson et al [8176530471116035]

April 11, 2014

Dear Dr Ursula D'Souza,

Thank you for the positive response and acceptance of the paper.
We have removed the reviewers’ names from the acknowledgments and have included the requested information. The manuscript is formatted according to the journal guidelines.
I trust everything is in order. Please let me know if you need anything else.
Thank you again for your assistance. We are very excited to have this paper published in your prestigious journal.

Sincerely,

Barbara Gibson, PhD, PT
Bloorview Childrens Hospital Foundation Chair in Childhood Disability Studies
Associate Professor, Department of Physical Therapy, University of Toronto
Senior Scientist, Bloorview Research Institute, Holland Bloorview Kids Rehabilitation Hospital